Author’s response to reviews

Title: Improving the outcome of infants born at less than 30 weeks of gestation - a randomized controlled trial of preventative care at home

Authors:

Alicia J Spittle (alicia.spittle@thewomens.org.au)
Carmel Ferretti (carmel.ferretti@mcri.edu.au)
Peter J Anderson (peterja@unimelb.edu.au)
Jane Orton (jane.orton@thewomens.org.au)
Abbey Eeles (a.eels@pgrad.unimelb.edu.au)
Lisa Bates (l.bates@pgrad.unimelb.edu.au)
Roslyn N Boyd (r.boyd@uq.edu.au)
Terrie E Inder (Inder_T@hids.wustl.edu)
Lex W Doyle (lwd@unimelb.edu.au)

Version: 2 Date: 27 October 2009

Author’s response to reviews:

Dear Editor,

The following points have been addressed:

- Please move the trial registration number from the Title page to the bottom of the Abstract.
  
  This has been changed.

- Please include a ‘Competing interests’ section between the Conclusions and Authors’ contributions. If there are none to declare, please write ’The authors declare that they have no competing interests’.
  
  This has been added

- Please include an Authors’ contributions section before the Acknowledgements and Reference list.
  
  This has been added

Yours Sincerely,

Alicia Spittle